Author's response to reviews

Title: A presentig with concha bullosa in another concha bullosa: a case report

Authors:

Ibrahim Cukurova (cukurova57@gmail.com)
Aytekin Yaz (aytekinyaz@hotmail.com)
Murat Gumussoy (mgumussoy@hotmail.com)
Orhan G Yigitbasi (gaziyigitbasi@yahoo.com)
Yucel Karaman (dr_ykaraman@hotmail.com)

Version: 10 Date: 5 December 2011

Author's response to reviews:

Author's response to reviews
Title: A concha bullosa inside another concha bullosa: a case report Authors:
Ibrahim Cukurova (cukurova57@gmail.com)
Aytekin Yaz (aytekinyaz@hotmail.com)
Murat Gumussoy (mgumussoy@hotmail.com)
Orhan G Yigitbasi (gaziyigitbasi@yahoo.com)
Yucel Karaman (dr_ykaraman@hotmail.com)

Version: 3 Date: 23 August 2011

Author's response to reviews: see over

Reviewer's report
Title: A concha bullosa inside another concha bullosa: a case report
Version: 2 Date: 4 July 2011
Reviewer: Lum Chee Lun
Reviewer #1 has no suggested changes for the manuscript.

Reviewer's report
Title: A concha bullosa inside another concha bullosa: a case report
Version: 2 Date: 4 August 2011
Reviewer: kin choo pua

Reviewer # 2
Comments to authors:
Interesting case,
Discussion was too long.

*Discussion was clipped. References reduced from fifteen to eleven. Extra cut of CT scan will be more comprehensive.
*One more Coronal paranasal computed tomography scan image added.

Quality of written English: Needs some language corrections before being published
*English spelling corrections were made according to the rules.